

Citation: Levecke B, Coffeng LE, Hanna C, Pullan RL, Gass KM (2021) Assessment of the required performance and the development of corresponding program decision rules for neglected tropical diseases diagnostic tests: Monitoring and evaluation of soil-transmitted helminthiasis control programs as a case study. PLoS Negl Trop Dis 15(9): e0009740. https://doi.org/10.1371/journal.pntd.0009740

Editor: Subash Babu, NIH-National Institute for Research in Tuberculosis-ICER, INDIA

Received: January 17, 2021

Accepted: August 17, 2021

Published: September 14, 2021

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

Funding: LEC gratefully acknowledges funding of the NTD Modelling Consortium by the Bill and Melinda Gates Foundation (OPP1184344), and funding from the Dutch Research Council (NWO, grant 016.Veni.178.023). KMG receives financial RESEARCH ARTICLE

Assessment of the required performance and the development of corresponding program decision rules for neglected tropical diseases diagnostic tests: Monitoring and evaluation of soil-transmitted helminthiasis control programs as a case study

Bruno Levecke^{1*}, Luc E. Coffeng², Christopher Hanna³, Rachel L. Pullan⁴, Katherine M. Gass⁵

 Department of Virology, Parasitology, Immunology and Physiology, Ghent University, Merelbeke, Belgium,
 Department of Public Health, Erasmus MC, University Medical Centre Rotterdam, Rotterdam, the Netherlands, 3 Global Project Partners, LLC, Oakland, California, United States of America, 4 Department of Disease Control, London School of Hygiene and Tropical Medicine, London, United Kingdom, 5 Neglected Tropical Diseases Support Centre, The Task Force for Global Health, Decatur, Georgia, United States of America

* bruno.levecke@ugent.be

Abstract

Recently, the World Health Organization established the Diagnostic Technical Advisory Group to identify and prioritize diagnostic needs for neglected tropical diseases, and to ultimately describe the minimal and ideal characteristics for new diagnostic tests (the so-called target product profiles (TPPs)). We developed two generic frameworks: one to explore and determine the required sensitivity (probability to correctly detect diseased persons) and specificity (probability to correctly detect persons free of disease), and another one to determine the corresponding samples sizes and the decision rules based on a multi-category lot quality assurance sampling (MC-LQAS) approach that accounts for imperfect tests. We applied both frameworks for monitoring and evaluation of soil-transmitted helminthiasis control programs. Our study indicates that specificity rather than sensitivity will become more important when the program approaches the endgame of elimination and that the requirements for both parameters are inversely correlated, resulting in multiple combinations of sensitivity and specificity that allow for reliable decision making. The MC-LQAS framework highlighted that improving diagnostic performance results in a smaller sample size for the same level of program decision making. In other words, the additional costs per diagnostic tests with improved diagnostic performance may be compensated by lower operational costs in the field. Based on our results we proposed the required minimal and ideal diagnostic sensitivity and specificity for diagnostic tests applied in monitoring and evaluating of soiltransmitted helminthiasis control programs.

support from the Coalition for Operational Research on Neglected Tropical Diseases (COR-NTD), which is funded at The Task Force for Global Health primarily by the Bill & Melinda Gates Foundation, by the UK aid from the British government, and by the United States Agency for International Development through its Neglected Tropical Diseases Program. CPH receives financial support from the Bill & Melinda Gates Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

The World Health Organization established an advisory group to identify and prioritize diagnostic needs for neglected tropical diseases, and to ultimately describe the minimal and ideal characteristics for new diagnostic tests. To support this advisory group, we developed two generic frameworks, which we applied to soil-transmitted helminthiases: one to explore and determine the required sensitivity (probability to correctly detect a diseased person) and specificity (probability to correctly detect a person free of disease), and another one to determine the corresponding samples size and decision rules during surveys. We showed that specificity rather than sensitivity will become more important when the program approaches the endgame of elimination and that the requirements for both parameters are inversely correlated, resulting in multiple combinations of sensitivity and specificity that allow for reliable decision making. We also highlighted that improving diagnostic performance results in smaller sample sizes for the same level of program decision making. In other words, the additional costs per diagnostic tests with improved diagnostic performance can be compensated by the lower operational costs in the field. Based on our results we proposed to the advisory group the required performance characteristics of diagnostic tests for soil-transmitted helminthiasis control programs.

Introduction

Recently, the Strategic and Technical Advisory Group (STAG), the principal advisory group to the World Health Organization (WHO) for the control of neglected tropical diseases (NTDs), decided that a single WHO working group was needed to help to identify and prioritize diagnostic needs [1]. One of the recommendations was that target product profiles (TPPs) for diagnostics were needed for soil-transmitted helminths (STHs) that would facilitate monitoring and evaluation of soil-transmitted helminthiasis control programs [2]. Generally, these TPPs describe the minimal and ideal characteristics, including but not limited to the sensitivity and the specificity (see [3] for previously published TPPs).

Soil-transmitted helminthiasis is a parasitic disease caused by a group of intestinal roundworms, including Ascaris lumbricoides (giant roundworm), Trichuris trichiura (whipworm), Ancylostoma duodenale and Necator americanus (hookworms). In 2019, it was estimated that they globally accounted for 1.97 million disability adjusted life years (12% of the total disease burden attributed to NTDs [4]). Given the route of STH transmission, infections and the associated disease burden predominantly occurs in (sub)tropical countries where transmission is facilitated by the optimal climate conditions for larval development, poverty, and lack of both sanitation and hygiene [4,5]. To fight the global STH-attributable morbidity, WHO recommends preventive chemotherapy (PC) programs, during which a single tablet of anthelmintic drugs (albendazole (400 mg) or mebendazole (500 mg)) is periodically administered to both pre-school and school age children and other at-risk populations living in endemic areas. The frequency of these large-scale deworming programs is based on whether the observed prevalence of STH infections (any species) exceeds a predefined program decision threshold. For example, at the start of the program it is recommended to distribute drugs twice a year when the prevalence is at least 50% and once a year when the prevalence is at least 20%. During the implementation phase, the prevalence of any STH infection is periodically re-evaluated to verify whether objectives are being met, and if necessary, to adjust the frequency of drug administration (prevalence \geq 50%: 3x PC / year; 50%> prevalence \geq 20%: maintain PC frequency;

20%> prevalence \geq 10%: 1x PC /year; 10%> prevalence \geq 2%: 1x PC/2 years; prevalence <2%: no PC) [6].

Traditionally, STHs have been diagnosed by detecting worm specific eggs in stool using a compound light microscope. Since the 1990s, Kato-Katz has been the WHO recommended diagnostic standard for quantifying eggs in stools [7], and hence it has been used to guide soiltransmitted helminthiasis control programs. During the last decade, a variety of new diagnostic tests have been introduced to the STH field, including both other microscopy-based [8-10], and DNA-based methods [11]. Each of these tests have important advantages and disadvantages over the Kato-Katz. Important advantages are a clearer microscopic view [8,9], a higher clinical sensitivity (referring to the proportion of diseased individuals correctly diagnosed as infected) [12,13], opportunities for automated egg counting and quality control [10,14], the ability to differentiate hookworm species [11] and to simultaneously detect parasites other than STHs [8,9,11]. The chief limitations of these novel tests are the need for well-equipped laboratories with well-trained technicians, the need to transport samples to a distant laboratory, the higher cost of processing large numbers of samples [15,16], and the lack of standardized protocols for DNA-based methods [11,17,18]. Currently, most diagnostic technologies based on biomarkers other than eggs or DNA (e.g. antigens, antibodies and metabolites) or other sample matrices (e.g. serum and urine) are either not yet explored or in research phase [19–22]. As these new diagnostic technologies transit from research to routine program tools, important consideration needs to be paid to the performance of these tools when used by NTD programs for making public health decisions.

In the present study, we developed a generic framework to explore the impact of diagnostic test sensitivity and specificity at the individual level on program decision making at the population level, with the ultimate aim to better define minimum TPP sensitivity and specificity targets for diagnostic tests for PC targeted NTDs. To this end, we first explored the impact of diagnostic sensitivity and specificity on the probability of making an incorrect program decision within a soil-transmitted helminthiasis control program: unnecessarily selecting a PC frequency that is greater than indicated by the true prevalence or prematurely reducing the frequency of PC. Subsequently, we developed a multi-category lot quality assurance sampling (MC-LQAS) framework that incorporates imperfect test performance to determine the corresponding sample size and associated decision rules.

Methods

Required sensitivity and specificity

General framework. A program decision is generally based on the outcome of an epidemiological survey in which N_{tot} subjects are screened for the presence of any infection. The observed prevalence (proportion of positive test results N^+ out of N_{tot} , which includes both false and true positive test results) is then compared to a program decision threshold (*T*). Rather than a proportion, one can also verify whether the number of positive test results N^+ exceeds *T'*. When we assume a diagnostic test *D* with a sensitivity of Se_d and a specificity Sp_d , a true underlying prevalence equal to $Prev_{true}$ and a sample size of N_{tot} , the probability observing at least *T'* positive results can be written as

$$P(N^{+} \geq T' | Prev_{true}, Se_d, Sp_d, N_{tot}) = \sum_{x=T'}^{N_{tot}} {N_{tot} \choose x} \cdot Prob_{+}^{x} \cdot (1 - Prob_{+})^{N_{tot} - x}$$
(1)

$$Prob_{+} = Se_{d} \cdot Prev_{true} + (1 - Sp_{d}) \cdot (1 - Prev_{true})$$
⁽²⁾

It is important to note that T' is not a fixed value, rather it will be a function of the total number of subjects screened (N_{tot}) , the program decision threshold (T) and the diagnostic performance of the test (Se_d and Sp_d), and this can be best illustrated with a few toy examples. Assume that we are screening 500 subjects (N_{tot}) with a perfect test ($Se_d = Sp_d = 100\%$) and the program decision threshold T is set at 50%, then T' equals 250. In case 1,000 subjects are screened with a perfect test, T' equals 500. Given the same N_{tot} (1,000 subjects) and diagnostic performance but a T of 2% instead of 50%, T' equals 20. When an imperfect diagnostic test ($Se_d = 80\%$ and $Sp_d = 80\%$) is used to screen 1,000 subjects and decisions are made around a program decision threshold T of 2%, T' equals 212 or more generally

$$T' = N_{tot} \cdot (Se_d \cdot T + (1 - Sp_d) \cdot (1 - T))$$
(3)

Combining (1)–(3) allows one to explore the impact of Se_d and Sp_d on the probability of making an incorrect program decision around a set of program decision thresholds *T*. For example, suppose 500 subjects (N_{tot}) are randomly selected from a population where the true underlying prevalence equals 45% ($Prev_{true}$) and a threshold of 50% (*T*) is used to make program decisions. The probability of $N^+ \ge T'$, and therefore unnecessarily selecting a PC frequency that is higher than indicated by the true prevalence, equals 1.4% when a perfect test ($Se_d = Sp_d = 100\%$) is applied and 9.7% for an imperfect test ($Se_d = Sp_d = 80\%$). Similarly, one can determine the probability of prematurely reducing the PC frequency. For example, if we change the true underlying prevalence from 45% to 55% ($Prev_{true} \ge T$), the probability of $N^+ < T'$, and therefore prematurely reducing the PC frequency equals 1.1% (= 1 – the probability of $N^+ \le T'$) when a perfect test ($Se_d = Sp_d = 80\%$).

Data generation. For this analysis, we fixed N_{tot} to 500, but varied both Se_d and Sp_d from 60% to 100% with 1% increments (resulting in 41 x 41 theoretic diagnostic tests) and $Prev_{true}$ from 0% to 100% with 0.2% increments. The program decision thresholds included the currently recommended thresholds for an STH control program (2%, 10%, 20% and 50%). In addition, we included program thresholds of 1% and 5%. This is because the current program thresholds are based on the observed prevalence using Kato-Katz thick smear, for which we know the specificity is not 100% [23,24]. As a consequence of this, the true underlying prevalence might be overestimated as it approaches zero.

Analysis of generated data. To further illustrate the interpretation of the obtained data, we worked out a toy example in Fig 1. This figure represents the probability of $N^+ \ge T'$ over a wide range of $Prev_{true}$ when an imperfect diagnostic test ($Se_d = Sp_d = 80\%$) was applied. Given a program decision threshold T of 50% (vertical straight line), we can deduce both the error related to unnecessarily selecting a PC frequency that is greater than needed ($\varepsilon_{overtreat}$) or prematurely reducing the frequency of PC ($\varepsilon_{undertreat}$). These errors are analogous to 1 minus the negative predictive value and 1 minus the positive predicted value, as used in recent NTD modelling studies on optimal program decision thresholds [25-27]. Subsequently, we can also deduce to what extent this diagnostic test allows for reliable decision making. In the present study, we will use two different operating definitions for 'reliable' based on both errors. In both definitions, we set the highest allowed probability of prematurely reducing frequency $(E_{undertreat})$ at 5%, whereas the highest allowed probability of falsely continuing or increasing PC frequency (Eovertreat) was set at either 10% and 25%. Generally, a lower value for Eundertreat is preferred as prematurely reducing PC frequency may lead to an increase in infection and morbidity. The two values for $E_{overtreat}$ allow to differentiate between both adequate ($E_{overtreat}$ = 25%) and ideal ($E_{overtreat} = 10\%$) program decision making scenarios. In the remainder of the document, we will refer to (in)adequate and (less than) ideal program decision making when



Fig 1. The general framework to determine the required sensitivity and specificity. The red line represents the probability (in %) of the number of positive test results (N^+) in a random sample of N_{tot} subjects (= 500) being at least T' (see Eq 3) based on an imperfect diagnostic test D (sensitivity (Se_d) = specificity (Sp_d) = 80%) over a wide range of true underlying prevalence ($Prev_{true}$). The vertical straight line represents the program decision threshold T of 50%. The yellow areas highlight the program errors $\varepsilon_{overtreat}$ ($Prev_{true} < 50\%$) and $\varepsilon_{undertreat}$ ($Prev_{true} \geq 50\%$). The horizontal black dashed lines represent a $\varepsilon_{overtreat}$ equal to 25% and a $\varepsilon_{undertreat}$ equal to 5% (= 100% - 95%), the vertical red dashed lines indicate the corresponding $Prev_{true}$. The grey zone indicates the range of $Prev_{true}$ for which the diagnostic test is considered inadequate to make a well-informed program decision ($\varepsilon_{overtreat} \geq 25\%$ and $\varepsilon_{undertreat} \geq 50\%$).

the $E_{overtreat}$ is set at 25% and 10% respectively. The values for $E_{undertreat}$ and $E_{overtreat}$ here have also been applied earlier to determine the sensitivity and specificity for diagnostic tests for other helminth diseases [28].

In the toy example (Fig 1), the diagnostic test performed at $\varepsilon_{undertreat} \leq 5\%$ when $Prev_{true}$ is at least 55.8% and at $\varepsilon_{overtreat} \leq 25\%$ when the $Prev_{true}$ is not higher than 47.2%. In other words, any program decision making within the $Prev_{true}$ interval] 47.2; 55.8 [is considered inadequate

when applying this test; we will refer to this interval as the 'grey zone'. It is expected that for a given sample size, the grey zone narrows with higher levels of sensitivity and specificity of diagnostic methods. Because the width of grey zones also depends on binomial variation, and thus on the program decision threshold itself, we quantified the grey zone for each combination Se_d and Sp_d and program decision threshold separately.

In order to further differentiate diagnostic tests with small grey zones from those with a wider zone, we classified the grey zone into three levels (level 1–3) for each program decision threshold *T* separately. This classification into 3 levels was based on the 25th and 75th percentile of the width of the grey zones (level 1: width of grey zone $< 25^{th}$ percentile; level 2: 75th percentile > width of grey zone $\ge 25^{th}$ percentile; level 3: width of grey zone $\ge 75^{th}$ percentile (see **S1 Table**) across all potential diagnostic methods that allowed for adequate program decision making. In other words, each of these diagnostic methods allowed for adequate decision making (*E*_{overtreat} is set at 25%) at a true underlying prevalence of zero and 100%. Finally, we arbitrarily classified the diagnostic tests into 'minimal' and 'optimal' based on their corresponding levels of grey zone across each of the 6 program decisions thresholds. Diagnostic performance was considered optimal when they resulted in level 1 grey zone for at least 3 out of the 6 program decision thresholds. In all other cases, the diagnostic test was considered 'minimal'.

MC-LQAS framework

General framework for LQAS. Lot quality assurance sampling (LQAS) is a technique to gather the minimal amount of information required for decision making, using a sample size as small as possible. Instead of constructing a precise estimate of a population parameter, LQAS aims to quantify whether the population parameter is above or below some decision cut-off *c* with some desired minimal probability. For STH, LQAS can be used to verify whether the observed number of positive test results (N^+) in a random sample (N_{tot}) equals or exceeds a predefined decision cut-off *c* [29,30], followed by continuing the current PC frequency if this is the case, and reducing the PC frequency in all other cases. The sample size N_{tot} and the corresponding decision cut-off *c* are chosen to satisfy two conditions. The first is that for some prevalence $Prev_{true}$ less than the program decision threshold T ($Prev_{true < T}$), the probability $\varepsilon_{overtreat}$ to select a PC frequency that is higher than indicated by the true underlying prevalence does not exceed the target probability $E_{overtreat}$. The second condition is that for some $Prev_{true}$ equal or above the program decision threshold T ($Prev_{true \geq T}$), the probability $\varepsilon_{undertreat}$ to prematurely reduce the PC frequency is not higher than $E_{undertreat}$. Based on Eqs (1)–(3) one can write these conditions as

$$P(N^{+} \ge c | Prev_{true < T}, Se_{d}, Sp_{d}, N_{tot}) = \sum_{x=c}^{N_{tot}} {N_{tot} \choose x} \cdot Prob_{+}^{x} \cdot (1 - Prob_{+})^{N_{tot} - x} \le E_{overtreat}$$
(4)

$$P(N^{+} < c | Prev_{true \geq T}, Se_{d}, Sp_{d}, N_{tot}) = \sum_{x=0}^{c-1} \binom{N_{tot}}{x} \cdot Prob_{+}^{x} \cdot (1 - Prob_{+})^{N_{tot} - x} \leq E_{undertreat}$$
(5)

where $Prob_+$ equals $Se_d \cdot Prev_{true < T} + (1 - Sp_d) \cdot (1 - Prev_{true < T})$ in (4) and $Se_d \cdot Prev_{true \ge T} + (1 - Sp_d) \cdot (1 - Prev_{true \ge T})$ in (5).

Process to determine the decision cut-off *c* **within LQAS.** Fig 2 further illustrates the process to determine the appropriate decision cut-off for two theoretical diagnostic tests. In this example, we determined the decision cut-off *c* for a sample size of 500 subjects (N_{tot}) that allowed for $E_{overtreat} \leq 25\%$ and $E_{undertreat} \leq 5\%$ at a $Prev_{true < T}$ arbitrarily set at 45% and at a $Prev_{true < T}$ arbitrarily set at 55% (program decision threshold T = 50%), respectively. To



Fig 2. The process to determine the decision cut-off c in a LQAs framework. The different panels in this figure illustrate the process to determine the decision cut-off c when 500 subjects (Ntot) are randomly recruited for both a perfect test (sensitivity (Se_d) = specificity (Sp_d) = 100%; **Panels A–C**) and an imperfect test ($Se_d = Sp_d = 80\%$); **Panels** D-F). Panels A and D represent the cumulative error of prematurely reducing the preventive chemotherapy (PC) $(\varepsilon_{undertreat})$ when the true underlying prevalence was arbitrarily set at 55% $(Prev_{true \geq T})$. The horizontal dashed line represents a $\varepsilon_{undertreat}$ of 5%, the red dashed line represents the allowed possible decision cut-off c resulting in a $\varepsilon_{undertreat} \leq 5\%$. The red area under the curve highlight all possible values for *c* resulting in a $\varepsilon_{undertreat} \leq 5\%$. Panels B and **E** represent the cumulative error of selecting a PC frequency that is higher than needed ($\varepsilon_{overtreat}$) when the true underlying prevalence was arbitrarily set at 45% (*Prev*_{true<1}). The horizontal dashed line represents a $\varepsilon_{overtreat}$ of 25%, the blue dashed line represents the lowest possible decision cut-off *c* resulting in a $\varepsilon_{overtreat}$ of $\leq 25\%$. The blue area under the curve highlights all possible values for c resulting in a $\varepsilon_{avertreat}$ of < 25%. Panels C and F represent the probability (in %) of the number of positive test results (N^+) in a random sample of N_{tot} subjects being at least c over a wide range of true underlying prevalence (Prev_{true}) based on the two extreme decision cut-offs (red line: lowest possible value; blue line: highest possible value). The vertical straight line represents the program decision threshold T of 50%. The horizontal black dashed lines represent a $\varepsilon_{overtreat}$ equal to 25% and a $\varepsilon_{undertreat}$ equal to 5% (= 100% -95%). The grey zone indicates the range of *Prev*_{true} for which decision making is inadequate ($\varepsilon_{overtreat}$ >25%) (blue dashed line) and $\varepsilon_{undertreat}$ >5% (red dashed line). In this example, the grey zone ranges from 45% to 55% by design.

contrast the findings, we determined *c* for both a perfect ($Se_d = Sp_d = 100\%$) and an imperfect test ($Se_d = Sp_d = 80\%$).

For both theoretical diagnostic tests there is a range of possible values for *c*. For a perfect test ($Se_d = Sp_d = 100\%$) any value between 233 (Fig 2B) and 257 (Fig 2A) can be used, whereas for an imperfect test ($Se_d = Sp_d = 80\%$) the range of possible values is narrower, only ranging from 244 (Fig 2E) to 247 (Fig 2D). This reduction in options of *c* for an imperfect test is also reflected in panels representing the probability the number of positive test results (N^+) in a random sample of N_{tot} subjects being at least *c* over a wide range of true underlying prevalence ($Prev_{true}$) (Fig 2C and 2F). Where both lines are almost overlapping for an imperfect test, there is a shift in $Prev_{true}$ of 5-point percent between both lines for a perfect test.

Expansion of framework to MC-LQAS. In STH control programs decisions are made around multiple program decision thresholds, and hence a MC-LQAS (based on multiple decision cut-offs) would be more appropriate. In 2012, Olives et al. described the mathematical underpinnings of a multi-category LQAS for schistosomiasis based on 2 decision cut-offs, resulting in three categories (three-way MC-LQAS) [31]. Fig 3 illustrates the built-up of a five-way MC-LQAS for program decisions around 4 program thresholds *T* currently used in STH



Fig 3. The build-up of multi-category LQAS for STH control program decision making using an imperfect test. The different panels illustrate the build-up of a multi-category LQAS around 4 program decision thresholds *T* (2%, 10%, 20% and 50%) when applying an imperfect test (sensitivity (Se_d) = 76% and specificity (Sp_d) = 99%) on 500 randomly selected subjects (N_{tot}). **Panel A** provides the provides the probability (in %) of the number of positive test results (N^+) in a random sample of N_{tot} subjects (= 500) being at least *c* separately for each of the 4 thresholds, their corresponding decision cut-offs ($c_{2\%} = 13$, $c_{10\%} = 41$, $c_{20\%} = 84$, $c_{50\%} = 182$) and true underlying prevalence $Prev_{true}$ ($Prev_{true < 2\%}$: 0.0%, $Prev_{true \geq 2\%}$: 4.0%; $Prev_{true < 10\%}$: 7.5%, $Prev_{true \geq 12}$. 5: 12.5%; $Prev_{true < 20\%}$: 15.0%, $Prev_{true \geq 20\%}$: 25.0%; $Prev_{true < 50\%}$: 45.0%, $Prev_{true \geq 50}$: 55.0%). Note that these $Prev_{true}$ values define the borders of the grey zone around the program thresholds and for these $Prev_{true}$ -values $\varepsilon_{overtreat} \leq 25\%$ and $\varepsilon_{undertreat} \leq 5\%$. The vertical straight line represents the program decision making is inadequate ($\varepsilon_{overtreat} \geq 25\%$ and $\varepsilon_{undertreat} > 5\%$). Panel **B** provides the same information as Panel A, but highlights the error of falsely scaling up the PC frequency (solid surfaces). **Panels C** and **D** represent the probability of correct program decision making across a wide range of $Prev_{true}$, where **Panel C** provides an overview of the relative contribution of $\varepsilon_{overtreat}$ (colored areas) in the program decision making.

programs ($T_1 = 2\%$, $T_2 = 10\%$, $T_3 = 20\%$ and $T_4 = 50\%$ [1]) when an imperfect test is used ($Se_d = 76\%$ and $Sp_d = 98\%$; this combination of Se_d and Sp_d allowed for accurate decision making (see **Table 1**). Fig **3A** provides the probability (in %) of the number of positive test results (N^+) in a random sample of N_{tot} subjects (= 500) being at least T' (see (**3**)) for each of the different thresholds, their corresponding decision cut-offs ($c_{2\%} = 13$, $c_{10\%} = 41$, $c_{20\%} = 84$, $c_{50\%} = 182$) and $Prev_{true}$ ($Prev_{true<2\%}$: 0.0%, $Prev_{true\geq2}$; 4.0%; $Prev_{true<10\%}$: 7.5%, $Prev_{true\geq10}$: 12.5%; $Prev_{true<20\%}$: 15.0%, $Prev_{true\geq20\%}$: 25.0%; $Prev_{true<50\%}$: 45.0%, $Prev_{true\geq50}$: 55.0%). Note that these $Prev_{true}$ -values define the borders of the grey zone around the program thresholds and for these $Prev_{true}$ -values for which $\varepsilon_{overtreat} \leq 25\%$ and $\varepsilon_{undertreat} \leq 5\%$. However, for a MC-LQAS we will need to consider the interaction between each of the 4 individual LQAS. For example, between 2 consecutive thresholds, there is not only the probability of prematurely reducing the PC frequency ($\varepsilon_{undertreat} = P(N^+ < c_{T_i} | T_i \leq Prev_{true} < T_{i+1}, Se_d, Sp_d, N_{tot})$) there is also the probability of falsely scaling up the PC frequency ($\varepsilon_{overtreat} = P(N^+ \geq c_{T_{i+1}} | T_i \leq Prev_{true} < T_{i+1}, Se_d, Sp_d, N_{tot})$). This $\varepsilon_{overtreat}$ around each of the program thresholds is highlighted in Fig **3B**. Combining both $\varepsilon_{undertreat}$ and

Table 1. The 207 diagnostic tests that allow for an adequate decision making. The table represents the width of the grey zone around the six program decision thresholds T (1%, 2%, 5%, 10%, 20% and 50%) that allowed for a sufficient decision making ($\varepsilon_{overtreat} \leq 25\%$ and $\varepsilon_{undertreat} \leq 5\%$) for each of the 207 pairs of sensitivity (Se_d) and specificity (Sp_d). For simplicity, we have classified the width of the grey zone into three levels (1–3) for each threshold and $\varepsilon_{undertreat}$ separately. This classification into 3 levels was based on the 25th and 75th percentile of the width of the grey zones (level 1: width of grey zone $<25^{th}$ percentile; level 2: 75th percentile > width of grey zone $\geq 25^{th}$ percentile; level 3: width of grey zone $\geq 75^{th}$ percentile (see S1 Table) across all potential diagnostic methods that allowed for adequate program decision making. In other words, each of these diagnostic methods allowed for adequate decision making ($\varepsilon_{overtreat}$ is set at 25%) at a true underlying prevalence of zero. Diagnostic tests were considered 'optimal' (blue) when they resulted in level 1 grey zone in at least 3 out of the 6 thresholds and did not result in a level 3 grey zone in any of the 6 program thresholds. In all other cases, the diagnostic test was considered 'minimal' (white).

Sp _d	Sed	Progra	m threshol	Type of test				
		50	20	10	5	2	1	
100	74-100	1	1	1	1	1	1	Optimal
	63-73	2	1	1	1	1	1	Optimal
	60-62	2	1	1	1	1	2	Optimal
99	75-100	1	1	1	1	1	2	Optimal
	60-74	2	1	1	1	1	2	Optimal
98	76-100	1	1	1	1	1	2	Optimal
	69–75	2	1	1	1	1	2	Optimal
	67–68	2	1	1	1	1	3	Minimal
	66	2	1	1	1	2	3	Minimal
	64-65	2	1	1	1	1	3	Minimal
	62-63	2	2	1	1	1	3	Minimal
97	77-100	1	1	1	1	1	2	Optimal
	72–76	2	1	1	1	1	3	Minimal
	68-71	2	1	1	1	2	3	Minimal
	63–67	2	2	1	1	2	3	Minimal
96	92-100	1	1	1	1	1	2	Optimal
	84-91	1	1	1	1	1	3	Minimal
95	98-100	1	1	1	1	1	2	Optimal
	93–97	1	1	1	1	1	3	Minimal
	87-92	1	1	1	1	2	3	Minimal
	85-86	1	1	1	1	1	3	Minimal
94	96-100	1	1	1	1	1	3	Minimal
	86-95	1	1	1	1	2	3	Minimal

https://doi.org/10.1371/journal.pntd.0009740.t001

 $\varepsilon_{overtreat}$ results into the probability of making incorrect program decisions, or in other words 1 -($\varepsilon_{undertreat}$ + $\varepsilon_{overtreat}$) or 1- ε provides the probability of correct program decision making. Fig 3C and 3D represent the probability of correct program decision making across a wide range of *Prev*_{true}, where Fig 3C provides an overview of the relative contribution of $\varepsilon_{undertreat}$ and $\varepsilon_{overtreat}$ in the program decision making. It is important to note that the different decision cut-offs c_{T_i} in this example are not based on (4) and (5) for each threshold separately, rather they were determined using the equations below

$$P(N^+ \ge c_{2\%} | Prev_{true<2\%}, Se_d, Sp_d, N_{tot}) \le E_1$$
(6)

$$P(c_{2\%} \le N^{+} < c_{10\%} | Prev_{true \ge 2\%}, Se_{d}, Sp_{d}, N_{tot}) \le 1 - E_{2} \&$$

$$P(c_{2\%} \le N^{+} < c_{10\%} | Prev_{true < 10\%}, Se_{d}, Sp_{d}, N_{tot}) \le 1 - E_{3}$$
(7)

$$P(c_{10\%} \le N^{+} < c_{20\%} | Prev_{true \ge 10\%}, Se_{d}, Sp_{d}, N_{tot}) \le 1 - E_{4} \& P(c_{10\%} \le N^{+} < c_{20\%} | Prev_{true < 20\%}, Se_{d}, Sp_{d}, N_{tot}) \le 1 - E_{5}$$

$$(8)$$

$$\begin{aligned} P(c_{20\%} \le N^{+} < c_{50\%} | Prev_{true \ge 20\%}, Se_{d}, Sp_{d}, N_{tot}) \le 1 - E_{6} \& \\ P(c_{20\%} < N^{+} < c_{50\%} | Prev_{true < 50\%}, Se_{d}, Sp_{d}, N_{tot}) \le 1 - E_{7} \end{aligned}$$
(9)

$$c_{20\%} \leq 17 \quad \langle c_{50\%} | 1707 true < 50\%, 500d, 50pd, 17tot \rangle \leq 1 \quad D_7$$

$$P(N^{+} < c_{50\%} | Prev_{true \ge 50\%}, Se_{d}, Sp_{d}, N_{tot}) \le E_{8}$$
(10)

where the *E* given $Prev_{true < T}$ (indicated with the odd subscript) represents the allowed probability of selecting a PC frequency that is greater than indicated by the true underlying prevalence, and those *E* given $Prev_{true \ge T}$ (indicated with an even subscript) represents the allowed probability of prematurely reducing the PC frequency. In this example, the *E* given $Prev_{true < T}$ was set at 25% and those given $Prev_{true > T}$ limit at 5%.

Determine sample size N_{tot} and decision cut-offs *c* for the required sensitivity and specificity within MC-LQAS. We will determine the sample size (N_{tot}) and the corresponding decision cut-offs c_{T_i} for those theoretical diagnostic tests that allowed for adequate or ideal program decision making. We varied the N_{tot} from 150–2,000 (by increments of 1), the corresponding decision cut-offs were based on (6)–(10). In this MC-LQAS, we considered all thresholds currently used in STH control programs (2%, 10%, 20% and 50%). For the corresponding $Prev_{true}$ limits, we used those used in the example illustrated in Fig 3 ($Prev_{true<2\%}$: 0.0%, $Prev_{true\geq2\%}$: 4.0%; $Prev_{true<10\%}$: 7.5%, $Prev_{true\geq10}$: 12.5%; $Prev_{true<20\%}$: 15.0%, Pre $v_{true\geq20\%}$: 25.0%; $Prev_{true<50\%}$: 45.0%, $Prev_{true\geq50}$: 55.0%). The *E* was set at 5% at $Prev_{true\geq T}$, *E* at $Prev_{true<T}$ was either set at 25% for adequate program decision making and at 10% for ideal program decision making.

Results

Required sensitivity and specificity

Figs 4 and **5** illustrate program decision making for a selection of the theoretic diagnostic tests, program decision thresholds and the level of reliable decision-making. **Fig 4** illustrates the program decision making for four theoretic distinct diagnostic tests (D_1-D_4) when decisions are made around the 50% program threshold. The diagnostic tests D_1-D_3 are imperfect diagnostic methods (**Fig 4A**: $Se_{d1} = Sp_{d1} = 60\%$; **Fig 4B**: $Se_{d2} = 100\%$ and $Sp_{d2} = 60\%$; **Fig 4C**: $Se_{d3} = 60\%$ and $Sp_{d3} = 100\%$), whereas D_4 is a perfect diagnostic method (**Fig 4D**: $Se_{d4} = Sp_{d4} = 100\%$). **Fig 5** contrasts the impact of (i) program decision errors ($E_{overtreat} = 25\%$ (**Fig 5A**) *vs*.



Fig 4. The program decision making around the 50% threshold for four theoretic diagnostic tests. The red line represents provides the probability (in %) of the number of positive test results (N^+) in a random sample of N_{tot} subjects (= 500) being at least T' (see Eq 3) using four theoretic distinct diagnostic tests (D_1-D_4). The diagnostic tests D_1-D_3 are imperfect diagnostic methods (**Panel A**: $Se_{d1} = Sp_{d1} = 60\%$; **Panel B**: $Se_{d2} = 100\%$ and $Sp_{d2} = 60\%$; **Panel C**: $Se_{d3} = 60\%$ and $Sp_{d3} = 100\%$), whereas D_4 is a perfect diagnostic method (**Panel D**: $Se_{d4} = Sp_{d4} = 100\%$). The grey area represents the range of true underlying prevalence for which program decision is inadequate ($\varepsilon_{overtreat} > 25\%$ and $\varepsilon_{undertreat} > 5\%$).

 $E_{overtreat} = 10\%$ (Fig 5B)), (ii) program decision thresholds (50% (Fig 5A) vs. 2% (Fig 5C) and (iii) diagnostic performance (diagnostic test D_2 (Fig 5C) vs. diagnostic test D_3 (Fig 5D)) on the grey zone.

Taken together, these figures highlight three important aspects. First, they indicate that program decision making becomes inadequate ($\varepsilon_{overtreat}$ >25% and $\varepsilon_{undertreat}$ >5%) when the true



Fig 5. The impact of program decision errors and diagnostic performance on the grey zone. The red line in each panel represents the probability (in %) of the number of positive test results (N^+) in a random sample of N_{tot} subjects (= 500) being at least T' (see Eq 3) (**Panels A** and **B**: T = 50%, **Panels C** and **D**: T = 2%) using 2 theoretic distinct imperfect diagnostic tests D_1 and D_2 ($Se_{d1} = 100\%$ and $Sp_{d1} = 60\%$ (**Panels A**, **B** and **C**); $Se_{d2} = 60\%$ and $Sp_{d2} = 100\%$ (**Panel D**)). The grey area represents the range of true underlying prevalence for which program decision is inadequate ($\varepsilon_{overtreat} > 25\%$ and $\varepsilon_{undertreat} > 5\%$ (**Panels A**, **D** and **C**) or not ideal ($\varepsilon_{overtreat} > 10\%$ and $\varepsilon_{undertreat} > 5\%$ (**Panel B**).

underlying prevalence (*Prev_{true}*) approaches the program decision threshold *T*, even if a perfect diagnostic method (D_4) is applied. Second, they confirm that improved diagnostic tests (Fig 4), less stringent program errors (Fig 5A and 5B) and lower program thresholds (Fig 5B and 5C) allow for narrower grey zones. Third, it is important to note that improving the specificity has a greater impact on the program decision making than improving the sensitivity, and that the impact of specificity increases as the program decision threshold shifts to 2%. Indeed, for a

program threshold of 50%, the grey zone of both diagnostic method D_2 ($Se_{d2} = 100\%$ and $Sp_{d2} = 60\%$) and D_3 ($Se_{d3} = 60\%$ and $Sp_{d3} = 100$) are equally wide (Fig 4), whereas for program decision threshold of one percent, the grey zone of diagnostic method D_3 is smaller compared to that one of diagnostic method D_2 (2%: ~3-point percent *vs.* ~8-point percent) (Fig 5C and 5D).

Fig 6 further summarizes the width of the grey zone for each of the 1,681 theoretic diagnostic tests by means of contour plots (each line represents the same width of grey zone) for adequate program decision making (**S1 Fig** provides the contour plots for ideal decision making). This figure highlights that multiple combinations of sensitivity and specificity can result in the



Fig 6. The width of grey zones around 6 program decision thresholds for 1,168 theoretic diagnostic tests. These contour plots illustrate the width of the grey zone for each of the unique combinations of sensitivity and specificity when decision making is adequate ($\varepsilon_{overtreat} \leq 25\%$ and $\varepsilon_{undertreat} \leq 5\%$), each line representing the same width of grey zone. The number beside the line represents the floor value of the width of the grey zone in % (e.g., any value $\geq 10\%$ and <11% is set at 10%).

https://doi.org/10.1371/journal.pntd.0009740.g006

same width of grey zone. For example, there are 408 combinations that result in a grey zone \sim 10-point percent wide around a program decision threshold *T* of 10%. However, for each of these combinations the sensitivity and specificity are inversely correlated (if sensitivity increases then the specificity decreases). Indeed, when the sensitivity is set at 60%, the specificity should not drop below \sim 83%. Similarly, a sensitivity of at least \sim 91% is required to obtain the same level of accurate decision making when the specificity is fixed at 60%. The figure also indicates that not all combinations can be recommended for monitoring and evaluating of STH programs, as the width of the grey zone would be too large to be relevant. An extreme case are the program decisions around a 2% threshold, where grey zones larger than 5-point percent would include a true underlying prevalence of zero, and hence would result in unnecessarily distributing drugs when disease has already been eliminated.

Of the 1,681 pairs of sensitivity (n = 41) and specificity (n = 41) that were evaluated, there were 207 combinations that allowed for adequate ($\varepsilon_{overtreat} \leq 25\%$ and $\varepsilon_{undertreat} \leq 5\%$) program decision making and 61 that resulted in ideal program decisions ($\varepsilon_{overtreat} \leq 10\%$ and $\varepsilon_{undertreat} \leq 5\%$) across each of the 6 program decision thresholds. In other words, they allowed for adequate or ideal decision making when the true underlying prevalence was zero and 100% across all thresholds. **Tables 1** and **2** provide an overview of the different possible diagnostic tests and their corresponding grey zone for $\varepsilon_{overtreat}$ less or equal to 25% and 10% respectively. For simplicity, we have classified the width of the grey zone into three levels (1–3) for each threshold separately. The classification into these 3 levels was based for each program decision threshold separately on the 25th and 75th percentile of the width of the grey zones (level 1: width of grey zone < 25th percentile; level 2: 75th percentile > width of grey zone $\geq 25^{th}$ percentile; level 3: width of grey zone $\geq 75^{th}$ percentile (see **S1 Table**).

Generally, each of these tables highlight four important aspects. First, they confirm that not all pairs of sensitivity and specificity allow for reliable decision making throughout all program phases. For example, combinations with specificity <94% are not included in Table 1. Second, they also confirm that diagnostic requirements become more stringent as program thresholds shift to 1%. This is because level 3 of the width of the grey zone in both tables is restricted by the program threshold of 1%. In other words, there are number of diagnostic tests that allowed for adequate or ideal program decision making around program decision thresholds between 2% and 50%, but failed to do so around a threshold *T* of 1%. Third, the requirements for both specificity and sensitivity are inversely correlated with each other; if the requirements are relaxed for one parameter, the requirements for the other one become more stringent for the other one. For example, if the specificity is 100% in Table 1, the lowest sensitivity to result in sufficient program decision making is 60%, whereas for a specificity of 94%, a sensitivity of at least 86% is required for sufficient decision making.

Fourth, when comparing Table 1 and Table 2 it becomes apparent that ideal program decisions require improved diagnostic tests. In contrast to an adequate program decision making (Table 1), for which there are 207 potential diagnostic tests, there are only 61 for ideal program decision making (Table 2). In addition, the requirements for specificity are more stringent. For an ideal decision making the specificity cannot drop below 99% (Table 2), whereas this was 94% across for an adequate decision making (Table 1).

In <u>Table 3</u> we cross tabulated the pairs of sensitivity and specificity across the two levels of program decision making (adequate *vs.* ideal) and two types of diagnostic test (minimal *vs.* optimal).

Sample size and decision cut-offs for the required sensitivity and specificity

Fig 7 summarizes the required sample size and the corresponding decision cut-offs c_i for the diagnostic tests summarized in **Table 3**. **Fig** 7**A** highlights that the required sample size

Table 2. The 61 diagnostic tests that allow for ideal decision making. The table represents the width of the grey zone around the six program decision thresholds T(1%, 2%, 5%, 10%, 20% and 50%) that allowed for a sufficient decision making ($\varepsilon_{overtreat} \leq 10\%$ and $\varepsilon_{undertreat} \leq 5\%$) for each of the 61 pairs of sensitivity (Se_d) and specificity (Sp_d). For simplicity, we have classified the width of the grey zone into three levels (1–3) for each threshold separately. This classification into 3 levels was based on the 25th and 75th percentile of the width of the grey zones (level 1: width of grey zone $< 25^{th}$ percentile; level 2: 75th percentile > width of grey zone $\ge 25^{th}$ percentile; level 3: width of grey zone $\ge 75^{th}$ percentile (see S1 Table) across all potential diagnostic methods that allowed for adequate program decision making. In other words, each of these diagnostic tests were considered 'optimal' (blue) when they resulted in level 1 grey zone around at least 3 out of the 6 thresholds and did not result in a level 3 grey zone in any of the 6 program thresholds. In all other cases, the diagnostic test was considered 'minimal' (white).

Sp _d	Sed	Program t	Type of test					
		50	20	10	5	2	1	
	96-100	1	1	1	1	1	1	Optimal
	81-95	2	1	1	1	1	1	Optimal
	70-80	2	2	1	1	1	1	Optimal
	61-69	2	2	1	1	1	2	Optimal
	60	3	2	1	1	1	2	Minimal
99	97-100	1	1	1	1	1	2	Optimal
	85-96	2	1	1	1	1	2	Optimal
	81-84	2	2	1	1	1	2	Optimal

https://doi.org/10.1371/journal.pntd.0009740.t002

decreases when the diagnostic performance improves. For example, where an imperfect diagnostic test ($Se_d = Sp_d = 96\%$) requires 301 subjects, this is only 200 for a perfect test ($Se_d = Sp_d = 100\%$). From the same panel we can deduce that improving the specificity has more impact on the sample size than improving sensitivity. For example, when improving the sensitivity from 96% to 100% when the specificity remains 96%, the sample size can only be reduced to 285, whereas improving the specificity from 96% to 100% when the sensitivity is fixed at 96%, the sample sizes can be further reduced to 209. Not unexpectedly, the sample size increases when an ideal rather than an adequate program decision making is required, and this is illustrated in Fig 7B. Fig 7C illustrates the variation in decision thresholds, highlighting that these values decrease when diagnostic tests become more perfect, which can be partially explained by the variation in sample size (see Fig 7A). The data used to determine the required diagnostic performance, the sample size and the corresponding decision cut-offs is provided S1 Data.

Discussion

This study presents a generic and readily adaptable framework to explore the impact of diagnostic test sensitivity and specificity at the individual level on program decision making, in this instance applied to STH decision thresholds. Our results emphasize that specificity—rather than sensitivity—will become increasingly important at the end-game as decision-relevant prevalence thresholds become lower. Although it is commonly stated that sensitivity is the most important diagnostic parameter when the prevalence drops [32–34], our study suggests the opposite. Indeed, the outcome of the simulation study indicated that there are fewer options for specificity (\geq 94%) than for sensitivity (\geq 60%), when it comes to sufficient program decision making, and that increasing specificity improved the overall accuracy of program decision making (narrower grey zones; **Fig 6**, **Tables 1** and **2 and S1 Fig**). Expanding this to explore the outcome of decision-making using MC-LQAS further highlighted that improving specificity would result in significantly less operational costs in the field (fewer subjects required to make adequate or ideal program decisions (**Fig 7**)).

Table 3. The diagnostic performance of minimal and optimal diagnostic tests for adequate and ideal decision making. Diagnostic tests were considered 'optimal' when they resulted in level 1 grey zone in at least 3 out of the 6 thresholds and did not result in a level 3 grey zone in any of the 6 program thresholds. In all other cases, the diagnostic test was considered 'minimal'. For simplicity, we have classified the width of the grey zone into three levels (1–3) for each threshold and $\varepsilon_{undertreat}$ separately. The classification into these 3 levels was based on the 25th and 75th percentile of the width of the grey zones (level 1: width of grey zone < 25th percentile; level 2: 75th percentile, width of grey zone $\geq 25^{th}$ percentile; level 3: width of grey zone $\geq 75^{th}$ percentile (see S1 Table)). For an adequate decision making the $\varepsilon_{overtreat} \leq 25\%$, whereas for ideal decision making this $\varepsilon_{overtreat} \leq 10\%$. For both levels of decision making $\varepsilon_{undertreat} \leq 5\%$.

		Program decision making							
		Adequate		Ideal					
		Specificity	Sensitivity	Specificity	Sensitivity				
Type of test	Minimal	98	62–68	100	60				
		97	63-76						
		96	84-91						
		95	85–97						
		94	86-100						
	Optimal	100	\geq 60	100	≥ 61				
		99	≥ 60	99	≥ 81				
		98	≥ 69						
		97	≥ 77						
		96	≥ 92						
		95	\geq 98						

https://doi.org/10.1371/journal.pntd.0009740.t003

Generally, our findings are very much in line with recent similar work [28]. In fact, these observations are not unexpected, and this can be best illustrated by an extreme case. Assume the disease is truly absent in population and samples are processed with an imperfect



Fig 7. The variation in sample size and decision cut-off for a selection of the diagnostic tests. Panel A describes the variation in sample size across varying sensitivity (60–100%) and specificity (94–100%) when program decision making is adequate. **Panel B** contrasts the sample size between adequate and ideal program decisions for two diagnostic tests (specificity = 99% and specificity = 100%). **Panel C** illustrates the variation in decision thresholds (2%-50%) across four diagnostic tests with the same specificity (96%) but varying sensitivity (64%, 70%, 80% and 90%).

https://doi.org/10.1371/journal.pntd.0009740.g007

diagnostic test, then the number of positive test results is determined by the specificity of the test only. For example, if we apply a test with a specificity of 95%, then there will be 5% (false) positive test results (Eq 2). Even if we have a true prevalence of 1% and a perfect sensitivity (100%), the majority of the positive tests will be false in nature.

Sensitivity and specificity need to be determined for each program use case

In the present study, we focused on defining the required specificity and sensitivity that allowed for adequate/ideal decision-making at each program treatment threshold. This strategy will result in diagnostic tests that can be used across all program decision thresholds; however, there may be diagnostic tests that perform well at a single threshold that are excluded by this approach (e.g., tests that perform well in high-prevalence settings). Indeed, all combinations of sensitivity and specificity allow for adequate and ideal program decisions around program thresholds of 20% and 50%. In other words, the required diagnostic performance will need to be determined for each program use case separately (see also Fig 6 and S1 Fig). For this, it will be equally important for the STH community to agree on the acceptable width of the grey zone separately for each program threshold, which in turn would provide a more justified criteria to classify diagnostic tests as 'optimal' and 'minimal' than those arbitrarily used in the present study.

Specificity and sensitivity are inversely correlated

Although the lowest possible specificity and sensitivity is 94% and 60% for adequate decision making and 99% and 60% for ideal program decision making (**Table 3**), it is important to note that the diagnostic requirements for specificity and sensitivity are inversely correlated. As a consequence of this, it would be inappropriate to independently report the lowest values of specific and sensitivity into a TPP, as this would lead to the development of diagnostic tests that result in poor program decision making. Rather, combinations/pairs of specificity and sensitivity will need to be incorporated. **S2 Table** lists the pairs of sensitivity and specificity that were eventually recommended to the STH subgroup. They include the pairs summarized in **Table 3**, excluding all combinations with a perfect sensitivity or specificity, because this was deemed unrealistic.

Currently used diagnostic methods may not allow for reliable decision making throughout an STH program

When comparing the recommended diagnostic performance (S2 Table) with the sensitivity and specificity for selection of currently available microscopic-based methods (e.g. direct smear, formol-ether concentration, Kato-Katz thick, McMaster, and (Mini-)FLOTAC) reported in a meta-analysis, it is clear that direct smear, formol-ether a single Kato-Katz and McMaster did not meet the requirements for detection of infections of any intensity for at least one of the three soil-transmitted helminths (Table 2 of [12]), and that in low endemic areas only FLOTAC would be a potential candidate (Table 3 of [12]). In a more recent study and assuming a perfect specificity [13], both a single and duplicate Kato-Katz, Mini-FLOTAC and qPCR did meet the required sensitivity for STH of any intensity (Table 3 of [13]), but when it concerns low intensity infections only qPCR remains as a potential candidate (Table 4 of [13]). FECPAK^{G2} did not meet any of the requirements. Although both studies indicate the potency of FLOTAC and qPCR, there are some important logistical obstacles to roll them out in large-scale deworming programs [16–18].

Extension of the (MC)-LQAS framework allows to both develop and compare program decision algorithms for imperfect tests

To our knowledge this is the first description of a five-way MC-LQAS framework that accounts for imperfects test. The expansion of this framework not only allows for developing program decision algorithms across imperfect tests, but can also be used to gain insights into the operational cost. For example, we showed that additional investments to improve the test (e.g., the specificity) may provide downstream benefits of reducing the required survey sample sizes for making adequate programme decisions. This is because diagnostic tests with improved specificity require smaller sample sizes for the same level of program decision making. In other words, any additional cost per diagnostic test with improved diagnostic performance can be compensated by savings in operational costs for testing in the field or laboratory. Therefore, it is recommended to split up operational costs for testing into the material cost per test and the number of tests that can be processed in an hour by one person in future cost-analyses. This level of costing detail would lead to greater evidence-based recommendations in the TPPs.

MC-LQAS framework needs to be adapted for 2-stage clustered sampling

In the current MC-LQAS framework we assumed that subjects are originating from the same cluster (e.g; community/school) and ignored the clustered nature of STH and assumed that these 500 subjects all represent one cluster (e.g. school/community). However, program decisions are not made at each cluster separately, rather decisions are made for a certain administrative or geographical area-the so-called implementation units-based on the aggregation of results across multiple clusters, with a number of subjects per cluster. In other words, programs employ 2-stage cluster sampling, whereby clusters are first chosen via random selection within an implementation unit and then a select number of subjects are chosen within each cluster. The development of a 2-stage cluster sampling MC-LQAS simulation approach was out of scope of the present study. A possible way forward would be to determine MC-LQAS around a 2-stage beta-binomial model, where the beta distribution describes the prevalence/ proportion of positive test results across clusters and the binomial distribution the proportion of positive test results within a cluster.

Both frameworks are generalizable to moderate-to-heavy intensity STH and any NTD program using population-based decision thresholds

Although the aforementioned frameworks were illustrated for program decision making around the prevalence of any STH infection, it is clear that both frameworks are agnostic to both the level of infection intensity and pathogen. For example, the results can also be used to make program decisions on whether the prevalence of moderate-to-heavy STH intensity infections has dropped below 2% [1]. Based on the diagnostic performance recommended in **S2 Table** and the recently reported probability of Mini-FLOTAC, McMaster and qPCR to correct classify moderate-to-heavy intensity infections when compared to Kato-Katz (Table 4 of [35]), we can deduce that only Mini-FLOTAC meets these requirements, though not for all STH species. Given that the schistosomiasis control programs use similar program decision thresholds [36], this framework will also provide insights for this NTD.

Supporting information

S1 Table. The thresholds to classify the width of the grey zone into three levels. This classification into 3 levels was based on the 25th and 75th percentile of the width of the grey zones

across all potential diagnostic methods for each program threshold *T* separately that allowed for an adequate program decision making (level 1: width of grey zone $< 25^{\text{th}}$ percentile; level 2: 75th percentile > width of grey zone $\ge 25^{\text{th}}$ percentile; level 3: width of grey zone $\ge 75^{\text{th}}$ percentile).

(DOCX)

S2 Table. The minimum and ideal sensitivity and specificity recommended by the STH subgroup.

(DOCX)

S1 Fig. The width of grey zones around 6 program decision thresholds for 1,168 theoretic diagnostic tests. These contour plots illustrate the width of the grey zone for each of the 1,168 unique combinations of sensitivity and specificity when decision making ideal ($\varepsilon_{overtreat} \leq 10\%$ and $\varepsilon_{undertreat} \leq 5\%$)each line represents the same width of grey zone. The number of the beside the line represents the floor value of the width of the grey zone in % (e.g., any value $\geq 10\%$ and <11% is set at 10%).

(TIF)

S1 Data. The data used to determine the required diagnostic performance, the sample size and the corresponding decision cut-offs.

(CSV)

Author Contributions

Conceptualization: Bruno Levecke, Luc E. Coffeng, Christopher Hanna, Rachel L. Pullan, Katherine M. Gass.

Formal analysis: Bruno Levecke, Luc E. Coffeng.

Investigation: Bruno Levecke, Luc E. Coffeng.

Methodology: Bruno Levecke, Luc E. Coffeng, Christopher Hanna, Rachel L. Pullan, Katherine M. Gass.

Validation: Bruno Levecke.

Visualization: Bruno Levecke, Luc E. Coffeng.

Writing - original draft: Bruno Levecke.

Writing – review & editing: Luc E. Coffeng, Christopher Hanna, Rachel L. Pullan, Katherine M. Gass.

References

- World Health Organization. Report of the first meeting of the WHO Diagnostic Technical Advisory Group for Neglected Tropical Diseases. Geneva, Switzerland: World Health Organization, 2020.
- Souza AA, Ducker C, Argaw D, King JD, Solomon AW, Biamonte MA, et al. Diagnostics and neglected tropical diseases roadmap: setting the agenda for 2030. Trans R Soc Trop Med Hyg. 2020; 0: 1–7.
- Lim MD, Brooker SJ, Belizario VY Jr, Gay-Andrieu F, Gilleard J, Levecke B, et al. Diagnostic tools for soil-transmitted helminths control and elimination programs: A pathway for diagnostic product development. PLoS Negl Trop Dis. 2018; 12: e0006213. https://doi.org/10.1371/journal.pntd.0006213 PMID: 29494581
- Vos T, Lim SS, Abbafati C, Abbas KM, Abbasi M, Abbasifard M, et al. Global burden of 369 diseases and injuries in 204 countries and territories, 1990–2019: a systematic analysis for the Global Burden of Disease Study 2019. Lancet. 2020; 396: 1204–1222. https://doi.org/10.1016/S0140-6736(20)30925-9 PMID: 33069326

- Jourdan PM, Lamberton PHL, Fenwick A, Addiss DG. Soil-transmitted helminth infections. Lancet. 2018; 391: 252–65. https://doi.org/10.1016/S0140-6736(17)31930-X PMID: 28882382
- 6. World Health Organization. 2030 targets for soil-transmitted helminthiases control programmes. Geneva, Switzerland: World Health Organization, 2020.
- World Health Organization. Basic laboratory methods in medical parasitology. Geneva, Switzerland. World Health Organization, 1991.
- Cringoli G, Rinaldi L, Maurelli MP, Utzinger. FLOTAC: new multivalent techniques for qualitative and quantitative copromicroscopic diagnosis of parasites in animals and humans. Nat Protoc. 2010; 5:503– 515. https://doi.org/10.1038/nprot.2009.235 PMID: 20203667
- Cringoli G, Maurelli MP, Levecke B, Bosco A, Vercruysse J, Utzinger J, et al. The Mini-FLOTAC technique for the diagnosis of helminth and protozoan infections in humans and animals. Nat Protoc. 2017; 12:1723–1732. https://doi.org/10.1038/nprot.2017.067 PMID: 28771238
- Ayana M, Vlaminck J, Cools P, Ame S, Albonico M, Dana D, et al. Modification and optimization of the FECPAK^{G2} protocol for the detection and quantification of soil-transmitted helminth eggs in human stool. PLoS Negl Trop Dis. 2018; 12: e0006655. https://doi.org/10.1371/journal.pntd.0006655 PMID: 30321180
- O'Connell EM, Nutman TB Molecular Diagnostics for soil-transmitted helminths. Am J Trop Med Hyg. 2016; 95: 508–513. https://doi.org/10.4269/ajtmh.16-0266 PMID: 27481053
- 12. Nikolay B, Brooker SJ, Pullan RL. Sensitivity of diagnostic tests for human soil-transmitted helminth infections: a meta-analysis in the absence of a true gold standard. Int J Parasitol. 2014; 44:765–74. https://doi.org/10.1016/j.ijpara.2014.05.009 PMID: 24992655
- Cools P, Vlaminck J, Albonico M, Ame S, Ayana M, Cringoli G, et al. Diagnostic performance of qPCR, Kato-Katz thick smear, Mini-FLOTAC and FECPAK^{G2} for the detection and quantification of soil-transmitted helminths in three endemic countries. PLoS Negl Trop Dis. 2019; 13:e0007446. https://doi.org/ 10.1371/journal.pntd.0007446 PMID: 31369558
- Vlaminck J, Cools C, Albonico M, Ame S, Ayana A et al. Therapeutic efficacy of albendazole against soil-transmitted helminthiasis in children measured by five diagnostic methods. PLoS Negl Trop Dis. 2019; 13: e0007471. https://doi.org/10.1371/journal.pntd.0007471 PMID: 31369562
- Levecke B, De Wilde N, Vandenhoute E, Vercruysse J. Field validity and feasibility of four techniques for the detection of *Trichuris* in simians: a model for monitoring drug efficacy in public health? PLoS Negl Trop Dis. 2009; 3:e366. https://doi.org/10.1371/journal.pntd.0000366 PMID: 19172171
- Speich B, Knopp S, Mohammed KA, Khamis IS, Rinaldi L, Cringoli G, et al. Comparative cost assessment of the Kato-Katz and FLOTAC techniques for soil-transmitted helminth diagnosis in epidemiological surveys. Parasit Vectors. 2010; 3:71. https://doi.org/10.1186/1756-3305-3-71 PMID: 20707931
- Papaiakovou M, Gasser RB, Littlewood DTJ. Quantitative PCR-Based Diagnosis of Soil-Transmitted Helminth Infections: Faecal or Fickle? Trends Parasitol. 2019; 35: 491–500. <u>https://doi.org/10.1016/j.pt.2019.04.006</u> PMID: 31126720
- Cools P, van Lieshout L, Koelewijn R, Addiss S, Ajjampur SSR, Ayana M, et al. First international external quality assessment scheme of nucleic acid amplification tests for the detection of *Schistosoma* and soil-transmitted helminths, including *Strongyloides*: A pilot study. PLoS Negl Trop Dis. 2020; 14: e0008231 https://doi.org/10.1371/journal.pntd.0008231 PMID: 32544158
- Vlaminck J, Supali T, Geldhof P, Hokke CH, Fischer PU, Weil GJ. Community rates of IgG4 antibodies to Ascaris haemoglobin reflect changes in community egg loads following mass drug administration. PLoS Negl Trop Dis 10: e0004532. https://doi.org/10.1371/journal.pntd.0004532 PMID: 26991326
- Lagatie O, Verheyen A, Van Asten S, Odiere MR, Djuardi Y, Levecke, et al. 2-Methyl-pentanoyl-carnitine (2-MPC): a urine biomarker for patent Ascaris lumbricoides infection. Sci Rep. 2020; 10: 15780. https://doi.org/10.1038/s41598-020-72804-y PMID: 32978457
- Lagatie O, Verheyen A, Van Hoof K, Lauwers D, Odiere MR, Vlaminck J, et al. Detection of Ascaris lumbricoides infection by ABA-1 coproantigen ELISA. PLoS Negl Trop Dis. 2020; 14: e0008807. <u>https://</u> doi.org/10.1371/journal.pntd.0008807 PMID: 33057357
- 22. Dana D, Vlaminck J, Ayana M, Tadege B, Mekonnen Z, Geldhofet al. Evaluation of copromicroscopy and serology to measure the exposure to *Ascaris* infections across age groups and to assess the impact of 3 years of biannual mass drug administration in Jimma Town, Ethiopia. PLoS Negl Trop Dis. 2020; 14: e0008037. https://doi.org/10.1371/journal.pntd.0008037 PMID: 32282815
- Speich B, Ali SM, Ame SM, Albonico M, Utzinger J, Keiser J. Quality control in the diagnosis of *Trichuris trichiura* and *Ascaris lumbricoides* using the Kato-Katz technique: experience from three randomised controlled trials. Parasit Vectors. 2015; 8:82. <u>https://doi.org/10.1186/s13071-015-0702-z</u> PMID: 25652120

- 24. Vlaminck J, Cools P, Albonico M, Ame S, Ayana M, Dana D, et al. An in-depth report of quality control on Kato-Katz and data entry in four clinical trials evaluating the efficacy of albendazole against soiltransmitted helminth infections. PLoS Negl Trop Dis. 2020; 14: e0008625. https://doi.org/10.1371/ journal.pntd.0008625 PMID: 32956390
- Coffeng LE, Le Rutte EA, Munoz J, Adams E, de Vlas SJ. Antibody and antigen prevalence as indicators of ongoing transmission or elimination of visceral leishmaniasis: a modeling Study. Clin Infect Dis. 2021; 72 (Suppl 3): S180–S187. https://doi.org/10.1093/cid/ciab210 PMID: 33906229
- 26. Farrell SH, Coffeng LE, Truscott JE, Werkman M, Toor J, de Vlas SJ, et al. Investigating the effectiveness of current and modified World Health Organization guidelines for the control of soil-transmitted helminth infections. Clin Infect Dis. 2018; 66 (suppl 4): S253–S259. https://doi.org/10.1093/cid/ciy002 PMID: 29860285
- Coffeng LE, Stolk WA, Golden A, de Los Santos T, Domingo GJ, de Vlas SJ. Predictive value of Ov16 antibody prevalence in different subpopulations for elimination of African onchocerciasis. Am J Epidemiol. 2019; 188(9): 1723–1732. https://doi.org/10.1093/aje/kwz109 PMID: 31062838
- Gass K (2020) Time for a diagnostic sea-change: Rethinking neglected tropical disease diagnostics to achieve elimination. PLoS Negl Trop Dis 14: e0008933. <u>https://doi.org/10.1371/journal.pntd.0008933</u> PMID: 33382694
- Addiss DG, Rheingans R, Twum-Danso NA, Richards FO. A framework for decision-making for mass distribution of Mectizan in areas endemic of Loa Ioa. Filaria J. 2003; 2 (Suppl. 1), S9. https://doi.org/10. 1186/1475-2883-2-S1-S9 PMID: 14975066
- Brooker SJ, Kabatereine NB, Myatt M, Stothard RJ, Fenwick A. Rapid assessment of *Schistosoma mansoni*: the validity, applicability and cost-effectiveness of the lot quality assurance sampling method in Uganda. Trop Med Int Health. 2005; 10: 647–658. <u>https://doi.org/10.1111/j.1365-3156.2005.01446.x</u> PMID: 15960703
- Olives C, Valadez JJ, Brooker SJ, Pagano M. Multiple category-lot quality assurance sampling: a new classification system with application to schistosomiasis control. PLoS Negl Trop Dis. 2012; 6: e1806. https://doi.org/10.1371/journal.pntd.0001806 PMID: 22970333
- Bergquist R, Johansen MV, Utzinger J. Diagnostic dilemmas in helminthology: what tools to use and when? Trends Parasitol. 2009; 25: 151–156. https://doi.org/10.1016/j.pt.2009.01.004 PMID: 19269899
- McCarthy JS, Lustigman S, Yang G-J, Barakat RM, García HH, Sripa B, et al. A Research Agenda for Helminth Diseases of Humans: Diagnostics for Control and Elimination Programmes. PLoS Negl Trop Dis. 2012; 6: e1601. https://doi.org/10.1371/journal.pntd.0001601 PMID: 22545166
- Mutombo PN, Man NWY, Nejsum P, Ricketson R, Gordon CA, Robertson G, et al. Diagnosis and drug resistance of human soil-transmitted helminth infections: A public health perspective. Adv Parasitol. 2019; 104:247–326. https://doi.org/10.1016/bs.apar.2019.02.004 PMID: 31030770
- Levecke B, Cools P, Albonico M, Ame S, Angebault C, Ayana M, et al. Identifying thresholds for classifying moderate-to-heavy soil-transmitted helminth intensity infections for FECPAK^{G2}, McMaster, Mini-FLOTAC and qPCR. PLoS Negl Trop Dis. 2020; 14: e0008296. https://doi.org/10.1371/journal.pntd. 0008296 PMID: 32614828
- World Health Organization. Accelerating work to overcome the global impact of neglected tropical diseases. A roadmap for implementation. Geneva: World Health Organization; 2012.